Chohan /28--Nov-05
Week 48 starting  28 Nov 05
SM18 Operation Meeting (Tuesdays)

Bench status on Monday 28 Nov 05:  
 11/28/2005 2:21:44 PM  
	6
	A1
	MBAR1289
	PT 12 Warm Up
	176.4

	11
	A2
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	291.33

	5
	B1
	SSS114
	PT 12 Warm Up
	300.28

	1
	B2
	SSS281
	PT 8 Fast De-excitation
	1.90

	7
	C1
	MBAL2133
	Prep 5 Cool Down
	96.91

	9
	C2
	MBBR2239
	ICS 4 Final connection
	289.99

	8
	D1
	SSS083
	Prep 3 CDW1 HV
	4.54 HV

	4
	D2
	SSS605
	PT 6.1 Training 1
	4.62 HV

	10
	E1
	MBAL3413
	ICS 5 Traveller Last
	294.1

	2
	E2
	MBBL2159
	PT 6.1 Training 1
	19.8

	3
	F1
	MBAR2247
	PT 11 4 K Quench SSL
	2.35

	12
	F2
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	288.48


Tempr. from TBSM18 site
 
   
Coordinator this week: vittorio; MTM support: 16-3602

 20 complete Magnet Tests cycles  13 Dipoles  + 7 SSS + 0 SSSS

 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight)
  [2201, 3388 2nd time ; 3139 3nd time ]
Situation From Cryogenics:  J-P Lamboy or ACR Rep.

Special Points:
· Why 4.4 K quenches on 2224( without shafts ) & Tempr problems
· Special SSS Q6 on D2 (nikolay, Marco)  SSW  issues @  weekend
AOB 
	Magnets under Test from 20-11-2005 to 26-11-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAR2226 
	[image: image5.png]



	


	20-11-2005 
	33 h.
	23 h.
	22 h.
	19 h.
	9 h.
	24-11-2005 
	106 h.
	Stripping
	
	

	2
	TBA1
	MBBL2201 
	[image: image7.png]



	


	14-11-2005 
	30 h.
	26 h.
	36 h.
	25 h.
	10 h.
	20-11-2005 
	127 h.
	Stripping
	
	

	3
	TBA2
	MBAL3378 
	[image: image9.png]



	


	16-11-2005 
	54 h.
	25 h.
	36 h.
	15 h.
	18 h.
	23-11-2005 
	148 h.
	Stripping
	
	

	4
	TBB1
	SSS149 
	[image: image11.png]



	


	20-11-2005 
	30 h.
	16 h.
	24 h.
	7 h.
	12 h.
	24-11-2005 
	89 h.
	Stripping
	
	

	5
	TBB1
	SSS161 
	[image: image13.png]



	


	16-11-2005 
	36 h.
	16 h.
	20 h.
	14 h.
	26 h.
	20-11-2005 
	112 h.
	Stripping
	
	

	6
	TBB2
	SSS147 
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	17-11-2005 
	36 h.
	28 h.
	14 h.
	11 h.
	21 h.
	22-11-2005 
	110 h.
	Stripping
	
	

	7
	TBB2
	SSS224 
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	22-11-2005 
	27 h.
	15 h.
	13 h.
	10 h.
	13 h.
	25-11-2005 
	78 h.
	Stripping
	
	

	8
	TBC1
	MBAR2249 
	[image: image19.png]



	


	17-11-2005 
	27 h.
	23 h.
	25 h.
	15 h.
	16 h.
	22-11-2005 
	106 h.
	Stripping
	
	

	9
	TBC1
	MBBL3410 
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	22-11-2005 
	22 h.
	23 h.
	15 h.
	19 h.
	23 h.
	26-11-2005 
	102 h.
	Stripping
	
	

	10
	TBC2
	MBAR2246 
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	19-11-2005 
	30 h.
	24 h.
	18 h.
	12 h.
	9 h.
	23-11-2005 
	93 h.
	Stripping
	
	

	11
	TBD1
	SSS165 
	[image: image25.png]



	


	15-11-2005 
	44 h.
	13 h.
	26 h.
	17 h.
	15 h.
	20-11-2005 
	115 h.
	Stripping
	
	

	12
	TBD1
	SSS208 
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	20-11-2005 
	34 h.
	17 h.
	23 h.
	7 h.
	28 h.
	25-11-2005 
	109 h.
	Stripping
	
	

	13
	TBD2
	SSS289 
	[image: image29.png]



	


	18-11-2005 
	25 h.
	16 h.
	18 h.
	8 h.
	12 h.
	21-11-2005 
	79 h.
	Stripping
	
	

	14
	TBE1
	MBAL1088 
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	19-11-2005 
	28 h.
	30 h.
	25 h.
	15 h.
	16 h.
	24-11-2005 
	114 h.
	Stripping
	
	

	15
	TBE2
	MBBR3341 
	[image: image33.png]



	


	17-11-2005 
	24 h.
	21 h.
	13 h.
	15 h.
	23 h.
	21-11-2005 
	96 h.
	Stripping
	
	

	16
	TBE2
	MBBR3388 
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	21-11-2005 
	33 h.
	20 h.
	28 h.
	14 h.
	15 h.
	26-11-2005 
	110 h.
	Stripping
	
	

	17
	TBF1
	MBAL3139 
	[image: image37.png]



	


	14-11-2005 
	33 h.
	28 h.
	37 h.
	19 h.
	20 h.
	20-11-2005 
	137 h.
	Stripping
	
	

	18
	TBF1
	MBBL1192 
	[image: image39.png]



	


	20-11-2005 
	25 h.
	24 h.
	17 h.
	13 h.
	9 h.
	24-11-2005 
	88 h.
	Stripping
	
	

	19
	TBF2
	MBAR1288 
	[image: image41.png]



	


	16-11-2005 
	27 h.
	21 h.
	13 h.
	25 h.
	20 h.
	21-11-2005 
	106 h.
	Stripping
	
	

	20
	TBF2
	MBBR3339 
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	21-11-2005 
	43 h.
	20 h.
	31 h.
	14 h.
	23 h.
	26-11-2005 
	131 h.
	Stripping
	
	


	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR2226
	12209.9 A.
(D2_u)
	12696.9 A.
(D1_L)
	12850 A.
[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL2201
	11958.2 A.
(D2-L-S01)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAL3378
	12336.3 A.
(D2_L_S02)
	12851 A.
[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS149
	11740.1 A.
(QI_34)
	12479.6 A.
(QI-12)
	12666 A.
(QI-34)
	[image: image48.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS161
	11617.6 A.
(QE 12)
	12217.7 A.
(QI 12)
	12388 A.
(QI-12)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS147
	12847.1 A.
(No Data)
	12850 A.
[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS224
	11153.6 A.
(QE-34)
	12059 A.
(QE-34)
	12652.1 A.
(QI-12)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAR2249
	9414 A.
(D2_L)
	12087 A.
(D1_U)
	12804.1 A.
(D1_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL3410
	11349 A.
(D2_L)
	12850 A.
[image: image53.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBAR2246
	11523.6 A.
(D2_U)
	11770.9 A.
(D1_L)
	12390.7 A.
(D1-U)
	12776 A.
(D1_U)
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	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS165
	12501.5 A.
(QE-12)
	12836.4 A.
(QI-12)
	12850 A.
[image: image55.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	SSS208
	12177.9 A.
(QE-34)
	12387.6 A.
(QE-34)
	12433.8 A.
(QE-34)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	SSS289
	11098 A.
(QI_34)
	12850 A.
[image: image57.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAL1088
	12296.7 A.
(D1-U)
	12539.5 A.
(D1-L)
	12777.7 A.
(D1_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBR3341
	9808.5 A.
(D2_U)
	10970.5 A.
(D2_L)
	12856 A.
[image: image59.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBBR3388
	9728 A.
(D1- L)
	9577.3 A.
(D1- U)
	11152.4 A.
(D2_U)
	11667 A.
(D1-U)
	12656.7 A.
(D1-U)
	12900 A.
[image: image60.jpg]



	-
	-
	-
	-
	-
	-
	-

	17
	MBAL3139
	12112.4 A.
(D2_L)
	12850 A.
[image: image61.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	18
	MBBL1192
	12529.6 A.
(D1-L)
	12648 A.
(D2-L)
	12850 A.
[image: image62.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	19
	MBAR1288
	12641.6 A.
(D2_L)
	12850 A.
[image: image63.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	20
	MBBR3339
	12728.4 A.
(D1-L)
	12850 A.
[image: image64.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
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	06Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_22_A
	TBE1


	Diode test on MB1088: Rcathode exceeded the limit of 15  (also wrong at cold).
	MTM
	
	Equip support called Pierre who asked to open a NC

	Tuesday_22_N
	TBF1


	MBBL1192: AQA alarm . MIITS AQA exceeds the limit 37(MIITS AQA = 45.1)in PT10 Min Energy.
Quench was out of the main coil.
	MTM
	
	Equip.support contacted during night. Why? TC contacted. He advised to do a SSL at 4.5 K. If the current is greater than 10200 A the magnet can be warmed up.

	Wednesday_23_M
	TBA1


	Loss of cryo OK when discharging the heaters after PT8.
	ACR
	
	

	Wednesday_23_A
	TBD1


	SSS208 :PT8 could no be launched. LF DAQ not getting armed
	AB/CO
	lf
	Equip Support called. CPU card error. Spare will be provided by Adrian.

	Wednesday_23_A
	TBB1


	SSS149 : Resistance value of the sextupole voltage taps does not match with the type of connection presented here but with the new connection type although this SSS is not belonging to the appropriate list.
	MTM
	
	Equip Support called. He will confirm the type of connection.

	Wednesday_23_N
	TBF


	Polarity switch blocked (not possible to go in right position)
	MTM
	
	Equip support called. Polarity inverter forced. 

	Thursday_24_M
	TBA1


	Lf DAQ not getting armed.
	AB/CO
	hf
	Guy solved the problem: CPU card not working, Card taken from cluster A has been taken.

	Thursday_24_N
	TBC


	Rotary switch blocked. Interlock present. 
	MTM
	
	Equip support called. IDCCT off.

Put on in the power supply.

	Friday_25_M
	TBD2


	Lyre test and HV tests done on SSS605. Electrical tests stopped because magnet scheme was missing and there is no traveller.
	MTM
	
	Traveller brought in the afternoon.

	Friday_25_A
	TBE


	LF DAQ is not working. 
	AB/CO
	lf
	The CPU card was moved in another bench. Problem fixed by Adrian.

	Friday_25_A
	TBD2


	IAP on SSS605 could not be launched. 
	MTM
	IAP
	Guy found that lf config files are not correct.  File were corrupted: need to be transferred again. 

	Friday_25_A
	TBC2


	MB1283: Provoked quench with LF heaters: AQA warning - Quench location outside main coil. Test repeated. Same result.

MB1283: Provoked quench with LF heaters: AQA did not detect the quench.

MB1283: Provoked quench with HF heaters: AQA indicated a quench location outside main coil.
	MTM
	AQA
	The AC compensation for D2-U-L was 30% more than what it is normally. The threshold was adjusted.

Equip support advised to perform HF provoked quench.

Equip support will check the results.

	Friday_25_N
	TBA2


	SPA because of cryo loss.
	ACR
	
	

	Friday_25_N
	TBD1


	SSS083 could not be connected by ICS because 3 wires cut on CFB side
	ACR
	
	Equip support called. Problem fixed. 

	Friday_25_N
	TBA2and C2


	Tests could not be launched because HWREC is not responding.
	IT
	
	Maryline called. Problem solved by Oracle expert (Oracle files corrupted in the DB).

	Saturday_26_A
	TBD2


	SSW stages are not correctly placed, i.e. stage A is on the CFB side and stage B is on the MRB side.
	MTM
	
	Nikolay asked Guy to installed properly the stages and give the correct sag_calibration_files for SSS605. Done.

	Saturday_26_N
	TBD2


	SSW at warm on SSS605. Movement in opposite direction for Y direction with respect to the command sent by the motor. 


	MTM
	
	DB file was copied from system 1 and did not correspond to the environment of system 3. 

Guy will correct the DB file.



	Saturday_26_N
	TBC2


	IAP @ Warm final on MB1283 not possible to launch because of a wrong temperature phase "Magnet @ 300K”. Temperature was below 300K but still above 250 K and ACR team did not give the OK
	ACR
	
	OK when the Temperature was closed to 300 K.

	Sunday_27_M
	TBA1


	PT4 on MB1289: quench at 780 A. Test repeated and same result. AQA did not detect a quench in the two times.
	MTM
	
	Maryline found that the AC compensation D1-D2 is not correct (offset above 100 mV). However no fault detected in IAP.

	Sunday_27_A
	TBF2


	HV test before powering on MB3406 : HV test with rack MY93 could not start. Same with MY 95 rack. Fault ; wrong cryo phase 

HV test could be launched but during the first step of test "CDW1" (dipole quench heaters vs. ground) the voltage was out of tolerance
	ACR

MTM
	
	Maryline called. She advised to re-start the PLCSE. In fact problem came fro cryo signal which cannot be delivered because magnet too cold (288K instead of 290 K).

Maryline checked and asked ICS to disconnect the magnet and re-do the WP04.

Test done after disconnection: problem remained. Decision is to remove the magnet from the bench and open the IFS box.

	Sunday_27_N
	All
	2 Roclas out of 3 are not working.
	
	
	

	Sunday_27_N
	TBF1


	MB2247: Quench below 12000 A  (IQ=11160 A) during ramp up for PT8 (SPA test at 12000 A).
	MTM
	
	Magnet detrained. Re-test needed with quench antenna.

	Sunday_27_N
	TBF1


	MB2247: Noise on direct pole voltages.
	MTM
	AQA
	No alarm: TC advises to go on.


